Author's response to reviews

Title: A needs-based method for estimating the behavioral health staff needs of community health centers

Authors:

   Bridget Teevan Burke (Bridget.Teevan@gmail.com)
   Benjamin F Miller (benjamin.miller@ucdenver.edu)
   Michelle Proser (Mproser@nachc.org)
   Stephen M Petterson (Speterson@aafp.org)
   Andrew W Bazemore (Abazemore@aafp.org)
   Eric Gopelerud (Goplerud-Eric@norc.org)
   Robert L Phillips Jr. (Bphillips@theabfm.org)

Version: 2 Date: 26 November 2012

Author's response to reviews:

Our manuscript was revised based on an email received outlining the specific need to: a) Add a title page with authors to the front of the manuscript; b) add a section about the availability of the data used for the paper; c) added the abstract back into the manuscript (it remains in the correct format according to your website); d) Added declaration of competing interests to the end of the manuscript; and, e) Added an author contributions section also at the end of the manuscript. All of these tasks have been done in the updated manuscript.